Author's response to reviews

Title: Social sciences research in neglected tropical diseases 2: A bibliographic analysis.

Authors:

Daniel D Reidpath (daniel.reidpath@monash.edu)
Pascale Allotey (pascale.allotey@monash.edu)
Subhash Pokhrel (subhash.pokhrel@brunel.ac.uk)

Version: 3 Date: 16 December 2010

Author's response to reviews:

I have made all Major Revisions:
1. Included Competing Interests
2. Included Authors' Contributions
3. Transferred sections of the Acknowledgments to the end of the Methods section, and updated the citations
4. Fixed all the references, except 113 (now 45) which was complete.

I have made all Minor Revisions:
Title Page: Qualifications removed, details of corresponding author included
Text: Coloured all black
Heading: 'Endnotes' nor reads 'Appendix'
Typography: Checked ... never my strong suit.